Author's response to reviews

Title: Decrease social inequalities return-to-work: development and design of a randomised controlled trial among women with breast cancer.

Authors:

Clémence Vidor (clemence.vidor@univ-lille3.fr)
Ariane Leroyer (ariane.leroyer@univ-lille2.fr)
Véronique Christophe (veronique.christophe@univ-lille3.fr)
Mélanie Seillier (melanie.seillier@carsat-nordpicardie.fr)
Jérome Foncel (jerome.foncel@univ-lille3.fr)
Jacques Bonneterre (j-bonneterre@o-lambret.fr)
Sophie Fantoni (fanquin@wanadoo.fr)

Version: 4 Date: 13 March 2014

Author's response to reviews:

- Submit the manuscript without additional files 1-4.
- We send the authorization of the CCP. It's the CPP NORTHWEST III (CAEN) gave a favorable opinion for the study APAPI the September 8, 2012. The CPP reference for this study is: 2012-13.